Author's response to reviews

Title: Spontaneous Corneal Melting in Pregnancy: A case report

Authors:

Sudesh K Arya (aryasudesh@yahoo.co.in)
Archana Malik (drarchanag2002@yahoo.com)
Sonika Gupta (sonika_gupta@yahoo.com)
Hemlata Gupta (hemlatagupta@rediffmail.com)
Sunandan Sood (soods32@yahoo.co.in)

Version: 6 Date: 26 October 2007

Author's response to reviews:

To,

Editor-in chief
Journal of Medical Case Reports

Subject: Submission of Revised Manuscript titled "Spontaneous Corneal Melting in Pregnancy": A case report.

Dear Sir,

Please find enclosed the revised version of manuscript titled "Spontaneous Corneal Melting in Pregnancy: A case report." Point wise explanation of reviewer's comments is given as follows:

Reviewer 2:
1. Corneal sensations were tested and were equally brisk in both the eyes.
Associate editor
1. Grammatical mistake corrected.
Within the text, the changes have been marked in red font for your convenience.
Kindly acknowledge the receipt of these documents.

Thanking you,

Yours sincerely

Archana Malik
Senior Resident
Department of Ophthalmology
Government Medical College & Hospital
Chandigarh, India